**Author's response to reviews**

**Title:** Clinical features, predictive factors and outcome of hyperglycaemic emergencies in a developing country.

**Authors:**

Anthonia O Ogbera (oogbera@yahoo.co.uk)  
Jacob Awobusuyi (awojaco@yahoo.com)  
Chioma Unachukwu (chiomaunachukwu@yahoo.com)  
Olufemi Fasanmade (oofasanmade@yahoo.com)

**Version:** 5  **Date:** 6 March 2009

**Author's response to reviews:**

We have included a title page with the authors’ names, affiliations and e mails addresses.  
The list of authors have been written as they appear in the submission pages  
A corresponding author has been indicated.  
Authors’ contributions have been indicated. That all authors read and approved of the manuscript has been indicated.  
Reference 10 has been properly written and is indicated below.  
The abstract has been included and matches that in the submission pages (This has been changed to match the present one). Azotaemia has been replaced with elevated urea levels.  
All keywords have been removed.  
Unnecessary page breaks and blank pages have been removed.